SECURITIES AND EXCHANGE COMMISSION

FORM S-8

Initial registration statement for securities to be offered to employees pursuant to employee benefit plans

Filing Date: **1996-09-16 SEC Accession No.** 0000950109-96-005999

(HTML Version on secdatabase.com)

FILER

GLATFELTER P H CO

CIK:41719| IRS No.: 230628360 | State of Incorp.:PA | Fiscal Year End: 1231

Type: S-8 | Act: 33 | File No.: 333-12089 | Film No.: 96630767

SIC: 2621 Paper mills

Mailing Address 228 S MAIN ST SPRING GROVE PA 17362 Business Address 228 S MAIN ST SPRING GROVE PA 17362 7172254711

SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM S-8 REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

P. H. GLATFELTER COMPANY

(Exact name of issuer as specified in its charter)

Pennsylvania 23-0628360 (State or other jurisdiction of (I.R.S. Employer incorporation or organization) Identification No.)

Spring Grove, Pennsylvania 17362 -----(Address of Principal Executive Offices) (zip Code)

> Profit Sharing Plan of P. H. Glatfelter Company and The Glatfelter Pulp Wood Company (Spring Grove Hourly Group) ______ (Full title of the plan)

> > Robert S. Wood, Secretary 228 S. Main Street Spring Grove, PA 17362

(717) 225-4711

(TELEPHONE NUMBER, INCLUDING AREA CODE, OF AGENT FOR SERVICE)

COPIES TO:

MORRIS CHESTON, JR., ESQUIRE
BALLARD SPAHR ANDREWS & INGERSOLL
1735 MARKET STREET - 51ST FLOOR
PHILADELPHIA, PENNSYLVANIA 19103

CALCULATION OF REGISTRATION FEE

<TABLE> <CAPTION>

Title of Securities to be Registered	Amount to be Registered (1)	Proposed Maximum Offering Price Per Share (2)	Proposed Maximum Aggregate Offering Price (2)	Amount of Registration Fee
<s> Common Stock \$.01 par val</s>		<c> \$17.875</c>	<c> \$12,959,375</c>	<c> \$4,468.75</c>

</TABLE>

- (1) Pursuant to Rule 416(a) under the Securities Act of 1933, this Registration Statement shall be deemed to cover an indeterminate number of additional shares of Common Stock issuable in the event the number of outstanding shares of the Company is increased by split-up, reclassification, stock dividend and the like.
- (2) Estimated solely for the purpose of computing the registration fee. In accordance with Securities and Exchange Commission Rule 457(c), the price shown is based on the average of the high and low price per share of Common Stock of the Company on September 11, 1996, \$17.875, as traded on the American Stock Exchange.

In addition, pursuant to Rule 416(c) under the Securities Act of 1933, this registration statement also covers an indeterminate amount of interests to be offered or sold pursuant to the employee benefit plan described herein.

INFORMATION REQUIRED IN THE SECTION 10(a) PROSPECTUS

The documents containing the information required to be included in Part I of this Registration Statement will be given or sent to all persons who are eligible to participate in the Profit Sharing Plan of P. H. Glatfelter Company and The Glatfelter Pulp Wood Company (Spring Grove Hourly Group), as amended effective January 1, 1989 and restated effective October 1, 1995, and as to be amended by Amendment No. 1 effective September 16, 1996 (the "Plan").

PART II

INFORMATION REQUIRED IN THE REGISTRATION STATEMENT

item 3. - INCORPORATION OF DOCUMENTS BY REFERENCE

The following documents filed with the Securities and Exchange Commission pursuant to the Securities Exchange Act of 1934 by P. H. Glatfelter Company (the "Company") (File No. 1-3560) or the Plan are incorporated herein by reference:

- (a) The Annual Report of the Company on Form 10-K for the year ended December 31, 1995.
- (b) The Quarterly Reports of the Company on Form 10-Q for the quarters ended March 31, 1996 and June 30, 1996.

Each document filed by the Company or the Plan subsequent to the date hereof pursuant to Sections 13(a), 13(c), 14 or 15(d) of the Securities Exchange Act of 1934, prior to the filing of a post-effective amendment which indicates that all securities offered hereby have been sold or which deregisters all securities then remaining unsold, shall be deemed to be incorporated by reference herein and to be a part hereof from the date of filing of such document.

2

Item 4.- DESCRIPTION OF SECURITIES

Common Stock

Holders of Common Stock, \$.01 par value, are entitled to one vote per

share and to vote cumulatively for directors. Therefore, each shareholder is entitled to cast as many votes in the election of directors as shall be equal to the number of shares of Common Stock held by such shareholder on the record date, multiplied by the number of directors to be elected. Under the Company's Bylaws, the directors are divided into three classes, two classes consisting of four directors each and one class consisting of three directors. As a result of the classified Board, a shareholder will need to own a greater number of shares in order to be assured of electing a director than would be necessary if the Board were not classified. Holders of Common Stock do not have pre-emptive rights.

The Common Stock is junior in all respects to the Company's Preferred Stock, par value \$50. The Common Stock is entitled to dividends as declared by the Board of Directors and may be repurchased by the Company, subject to the satisfaction of dividend and sinking fund requirements with respect to outstanding Preferred Stock. There are currently no outstanding shares of Preferred Stock.

The Transfer Agent and Registrar for the Common Stock is currently American Stock Transfer & Trust Company. The Common Stock of the Company is currently traded on the American Stock Exchange.

On August 12, 1996, there were 54,361,980 authorized shares of Common Stock of the Company and 42,651,128 outstanding shares of Common Stock.

Item 5.- INTERESTS OF NAMED EXPERTS AND COUNSEL

Not applicable.

Item 6.- INDEMNIFICATION OF DIRECTORS AND OFFICERS

See Section 2.11 and Sections 3.1 through 3.7 of the Company's Bylaws and Section 1713 and Sections 1741 through 1750 of the Pennsylvania Business Corporation Law of 1988.

The Company's Bylaws include a provision to eliminate the personal liability of its directors for monetary damages for breach or alleged breach of their duty of care to the full extent permitted by Pennsylvania law. In addition, the Company's Bylaws provide that the Company shall indemnify its directors and officers to the full extent permitted by Pennsylvania law.

3

The Company has insurance coverage for losses by any person who is or hereafter may be a director or officer of the Company arising from claims against that person for any wrongful act (subject to certain exceptions) in his capacity as a director or officer of the Company. The policy also provides for

reimbursement to the Company for indemnification given by the Company, pursuant to common or statutory law or its Articles of Incorporation or Bylaws to any such person arising from any such claim. The policy's coverage is limited to a maximum of \$20,000,000 for each loss and each policy year and there is a deductible of \$1,750,000 for the Company.

Item 7.- EXEMPTION FROM REGISTRATION CLAIMED

Not applicable.

Item 8.- EXHIBITS

4.1 Specimen copy of Common Stock Certificate (incorporated by reference to Exhibit 4 to the Company's Registration Statement on Form S-8, Reg.

No. 33-54409)

- 4.2 Articles of Incorporation as amended through January 26, 1994 (restated for the purpose of filing on EDGAR) (incorporated by reference to Exhibit 3(c) to the Company's Form 10-K for the year ended December 31, 1993)
- 4.3 Profit Sharing Plan of P. H. Glatfelter Company and The Glatfelter Pulp Wood Company (Spring Grove Hourly Group), as amended by Amendment No. 1 effective September 16, 1996
- 5.1 Opinion of Ballard Spahr Andrews & Ingersoll re legality
- 5.2 Copy of the Internal Revenue Service determination letter, dated September 18, 1995
- 5.3 Opinion of Ballard Spahr Andrews and Ingersoll re compliance with the Employee Retirement Income Security Act of 1974 ("ERISA") (included in Exhibit 5.1)
- 15 Letter re unaudited interim financial information
- 23.1 Consent of Deloitte & Touche LLP
- 23.2 Consent of Ballard Spahr Andrews & Ingersoll (included in Exhibit 5.1)
- 24 Power of Attorney (included on signature page)

4

Item 9.- UNDERTAKINGS

A. Rule 415 Offering

The undersigned Registrant hereby undertakes:

- (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this registration statement:
 - (i) To include any prospectus required by section 10(a)(3) of the Securities Act of 1933;
 - (ii) To reflect in the prospectus any facts or events arising after the effective date of the registration statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the registration statement. Notwithstanding the foregoing, any increase or decrease in the volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus filed with the Commission pursuant to Rule 424(b) if, in the aggregate, the changes in volume and price represent no more than a 20% change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in the effective registration statement;
 - (iii) To include any material information with respect to the plan of distribution not previously disclosed in the registration statement or any material change in such information in the registration statement;

Provided, however, that paragraphs (A)(1)(i) and (A)(1)(ii)

do not apply if the information required to be included in a post-effective amendment by those paragraphs is contained in periodic reports filed with or furnished to the Commission by the Registrant pursuant to section 13 or section 15(d) of the Securities Exchange Act of 1934 that are incorporated by reference in the registration statement.

(2) That, for the purpose of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and

5

the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

(3) To remove from any registration by means of a post-effective

amendment any of the securities being registered which remain unsold at the termination of the offering.

B. Filings Incorporating Subsequent Exchange Act Documents By Reference

The undersigned Registrant hereby undertakes that, for purposes of determining any liability under the Securities Act of 1933, each filing of the Registrant's annual report pursuant to section 13(a) or section 15(d) of the Securities Exchange Act of 1934 (and, where applicable, each filing of an employee benefit plan's annual report pursuant to section 15(d) of the Securities Exchange Act of 1934) that is incorporated by reference in the registration statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

C. Request for Acceleration of Effective Date or Filing of Registration Statement on Form S-8

Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of the Registrant pursuant to the foregoing provisions, or otherwise, the Registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the Registrant of expenses incurred or paid by a director, officer or controlling person of the Registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the Registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.

6

SIGNATURES

The Registrant. Pursuant to the requirements of the Securities Act of

1933, the Registrant certifies that it has reasonable grounds to believe that it meets all of the requirements for filing on Form S-8 and has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized in the City of Spring Grove, Commonwealth of Pennsylvania, on the date written below.

P. H. GLATFELTER COMPANY

September 16, 1996

By: /s/ R. P. Newcomer

R. P. Newcomer
Senior Vice President,
Treasurer and Chief
Financial Officer

POWER OF ATTORNEY

KNOW ALL MEN BY THESE PRESENTS, that each person whose signature appears below constitutes and appoints R. P. Newcomer and R. S. Wood and each of them, as true and lawful attorneys-in-fact and agents with full power of substitution and resubstitution, for him and in his name, place and stead, in any and all capacities, to sign the Registration Statement and to file the same, with all exhibits thereto, and other documents in connection therewith, with the Securities and Exchange Commission, granting unto said attorneys-in-fact and agents full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes as he might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents, or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, this Registration Statement has been signed by the following persons in the capacities and on the dates indicated.

7

The Plan. Pursuant to the requirements of the Securities Act of 1933,

the trustees (or other persons who administer the employee benefit plan) have duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized in the City of Spring Grove, Commonwealth of Pennsylvania, on the date written below.

September 16, 1996

P. H. GLATFELTER COMPANY
EMPLOYEE BENEFIT COMMITTEE

By: /s/ R. P. Newcomer

R. P. Newcomer Chairman

<table> <caption> Signature</caption></table>	Title	Date	
<\$>	<c></c>	<c></c>	
P. H. Glatfelter III	Chairman Emeritus and Director		
/s/ T. C. Norris T. C. Norris		September	16, 1996
/s/ R. P. Newcomer R. P. Newcomer		September	16, 1996
/s/ C. M. Smith C. M. Smith	Comptroller	September	16, 1996
N. DeBenedictis	Director		
/s/ R. E. ChappellR. E. Chappell	Director	September	16, 1996
/s/ G. H. Glatfelter	Director	September	16, 1996

/s/ G. H. Glatfelter II	Director		September	16,	1996
G. H. Glatfelter II					
/s/ R. S. Hillas	Director		September	16,	1996
R. S. Hillas					

							9			
/s/ M. A. Johnson II			September	16,	1996					
M. A. Johnson II										
/s/ P. R. Roedel	Director		September	16,	1996					
P. R. Roedel										
/s/ J. M. Sanzo	Director		September	16,	1996					
J. M. Sanzo										
/s/ R. L. Smoot	Director		September	16,	1996					
R. L. Smoot										
		10								

EXHIBIT INDEX

EXHIBIT NO.

DESCRIPTION

4.3	Profit Sharing Plan of P. H. Glatfelter Company and The Glatfelter Pulp Wood Company (Spring Grove Hourly Group), as amended by Amendment No. 1 effective September 16, 1996
5.1	Opinion of Ballard Spahr Andrews & Ingersoll re legality
5.2	Copy of the Internal Revenue Service determination letter, dated September 18, 1995
5.3	Opinion of Ballard Spahr Andrews and Ingersoll re compliance with the ERISA (included in Exhibit 5.1)
15	Letter re unaudited interim financial information
23.1	Consent of Deloitte & Touche LLP
23.2	Consent of Ballard Spahr Andrews & Ingersoll (included in Exhibit 5.1)

EXHIBIT 4.3

PROFIT SHARING PLAN OF

P.H. GLATFELTER COMPANY AND THE GLATFELTER PULP WOOD COMPANY (SPRING GROVE HOURLY GROUP)

AS AMENDED, EFFECTIVE JANUARY 1, 1989 RESTATED EFFECTIVE OCTOBER 1, 1995

The Employees' Profit Sharing Plan of P.H. Glatfelter Company and The Glatfelter Pulp Wood Company was originally effective January 1, 1961. Effective January 1, 1980, that Plan was superseded by two new plans, one of which was the Hourly Employees' Profit Sharing Plan of P.H. Glatfelter Company and The Glatfelter Pulp Wood Company. This Plan is a continuation of that Plan.

TABLE OF CONTENTS

<TAB <CAP

BLE> PTION>		
FILON		PAGE
	SECTION 1 DEFINITIONS AND CONSTRUCTION	
<s> 1.1 1.2 1.3</s>	CC> Definitions Uncapitalized Terms Gender	13
	SECTION 2 ELIGIBILITY	
2.1 2.2 2.3 2.4	Active Members as of December 31, 1988 Other Eligible Employees	14 14

SECTION 3 COMPANY CONTRIBUTIONS

3.1 3.2 3.3 3.4

 P.H. Glatfelter Company The Glatfelter Pulp Wood Company Hourly Spring Grove Group Percentage Nonforfeitability | 16 17 || | SECTION 4 ALLOCATIONS TO SHARING MEMBERS | |
4.1	Allocation	_
4.2	Deferred Elections	
4.3	Limitation on Elections	21
4.4	Payment of Contributions	
4.5	Cash Payments	
4.6	Maximum Allocations	
4.7	Predecessor Plan	36
	SECTION 5	
	ADMINISTRATION OF THE FUND	
5.1	Investment by Trustee	
5.2	Investment Portfolios	
5.3	Valuation of Accounts	3 /
	SECTION 6	
	BENEFITS FOR MEMBERS	
6.1	Separation from Service or Death	38
6.2	Emergency Withdrawals	
6.3	Other Withdrawals	42
SECTION 7 METHOD OF DISTRIBUTION OF BENEFITS

	7.1 7.2 7.3 7.4 7.5 7.6 7.7	Form of Distribution. Death Benefits. Life Expectancy. Minimum Distribution Requirements. Applicability. Transfer of Account to Other Plan. Transfers to Salaried Group.	48 51 51 51 52
		SECTION 8	
		ADMINISTRATION	
	8.1	Administrator	
	8.2	Named Fiduciary	
	8.3	Rules of Administration	
	8.4 8.5	Services to the Plan	
	8.6	Claims Procedure	
	8.7	Liability for Administration	
<td>E></td> <td>-</td> <td></td>	E>	-	
		ii 08/1	13/96
		SECTION 9	
		AMENDMENT AND TERMINATION	
TABLE	'>		
CAPTI			
	<s></s>	<c></c>	<c></c>
	9.1	Amendment	59
		GEGETON 10	
		SECTION 10 GENERAL PROVISIONS	
	10.1	Investment Expenses	60
	10.2	Other Expenses	
	10.3	Voting Rights	
	10.4 10.5	Annual Statement	
	10.5	Payments Due Minors or Incapacitated Persons	
	10.7	Uniform Administration	
	10.8	Return of Nondeductible or Erroneous	
		Contribution	61
	10.9	Source of Payment	62
	10.10	No Right to Employment	
	10.11	Inalienability of Benefits	62

10.12	Mergers, etc	63
	Governing Law	
10.14	Information Required of Participants	63
10.15	Missing Persons	63
10.16	Effective Date	64

 | |iii

08/13/96

SECTION 1

DEFINITIONS AND CONSTRUCTION

1.1 Definitions. Whenever used in this Plan:

"Account" means, for each Member, an account established to ----record his interest in the Fund.

"Account Balance" means, for each Member, the net amount -----credited to his Accounts at any given time.

"Active Member" means an individual who has become an Active
----Member as provided in Section 2 and has at all times thereafter been an Eligible
Employee.

"Administrator" means the administrator of the Plan as -----designated under Section 8.1.

"Affiliate" means any trade or business (other than the

Company) that is a member of a controlled group of corporations, a group of commonly controlled trades or businesses, or an affiliated service group (as determined under Sections 414(b), (c) or (m) of the Code) of which the Company is a member, or is required to be aggregated with the Company pursuant to regulations under Section 414(o) of the Code; provided, however, that for purposes of Section 4.6 hereof, such determination is modified in the manner set forth in Section 415(h) of the Code.

"Age" means age on last birthday, except that an individual --- attains age 70-1/2 on the corresponding date in the sixth calendar month following the month in which his seventieth

birthday falls (or the last day of such sixth month if there is no such corresponding date therein).

"Beneficiary" means, for each Member, the individual or entity ------described in Section 7.2.

"Benefit Accrual Period" means, with respect to any Active

Member, the portion of a Plan Year that begins on the later of January 1 or the date on which he becomes an Active Member and ends on the earliest of December 31, the date on which he ceases to be an Eligible Employee but continues to be an Employee, the date he actually retires from the Company, or the date he dies.

"Board" means the Board of Directors of P.H. Glatfelter

Company.

"Code" means the Internal Revenue Code of 1986, as amended (or ---- any predecessor or successor thereto).

"Company" means P.H. Glatfelter Company, a Pennsylvania

corporation, The Glatfelter Pulp Wood Company, a Maryland corporation, and/or Spring Grove Water Company, a Pennsylvania corporation, as the context may indicate.

"Earnings" means, with respect to any Employee, for any Plan

Year, all straight time earnings received during his Benefit Accrual Period of the Plan Year, including shift differential, vacation pay and holiday pay, but excluding premium for overtime, moving expenses, severance pay, suggestion awards, amounts paid or credited as profit sharing distributions, and taxable compensation attributable to matching contributions under

2 08/13/96

the P.H. Glatfelter Company Employee Stock Purchase Plan. With respect to any Plan Year, the "Earnings" of each Employee shall not include that portion of his Earnings in excess of the amount to which the \$200,000 limit (for Plan Years beginning before January 1, 1994) or the \$150,000 (for Plan Years beginning on or after January 1, 1994) limit of Code section 401(a)(17) has been indexed. In determining the Earnings of an Employee for purposes of applying the limitation of Code section 401(a)(17), the rules of Code section 414(q)(6) shall apply, except that the term "family members" shall include only the spouse of the Employee and any lineal descendants who have not attained Age 19 before the

close of the Plan Year. In applying the rules of section 414(q)(6) of the Code, the limitation of Code section 401(a)(17) shall be allocated among family members in proportion to their Earnings without regard to such limitation.

"Effective Date" of the Plan means January 1, 1961.

"Elective Contribution" means, for any Plan Year, for each

Sharing Member, 100% of the Company contribution allocated to him under Section 4.1(a) for that year minus any Nonelective Contribution for the year, but not to exceed \$7,000 (or such other amount as may be fixed for elective deferrals by the Secretary of the Treasury or his delegate). This amount shall be the amount of the Company contribution that the Sharing Member may elect to receive in cash or to defer under the Plan.

08/13/96

"Eligibility Computation Period" means with respect to any

Employee, the twelve-month period beginning on his Employment Date and all calendar years after his Employment Date.

3

"Eligible Employee" means an Employee who is an hourly employee

of the Company, in its Spring Grove Group; except that any Employee with respect to whom retirement benefits have been the subject of good faith collective bargaining shall not be an Eligible Employee unless the collective bargaining agreement relating to his employment so provides, and, effective October 1, 1995, Employees in the Conference Group shall not be Eligible Employees.

"Employee" means any individual employed by the Company or an

Affiliate or absent under circumstances included in his Full-Time Employment. An individual shall be deemed to be employed by the Company or an Affiliate if such individual is employed directly by the Company or such Affiliate or is a leased employee, within the meaning of Section 414(n) or 414(o) of the Code, with respect to whose services the Company or such Affiliate is the recipient and to whom Section 414(n)(5) of the Code does not apply. An individual shall cease to be an Employee upon his resignation, discharge, death or actual retirement from the Company.

"Employment Date" means the day on which an Employee completes

his first Hour of Service.

"Full-Time Employment" means employment by the Company in a

position designated by the Company as full-time.

Any of the following types of absence shall be counted as Full-Time Employment if it immediately follows a period of Full-Time Employment:

- (a) absence or leave for sickness, accident, government service or extended vacation approved by the Company pursuant to uniform policy equally applicable to all Employees.
- (b) layoff of less than one year, if the Employee returns to work within one week after being recalled by certified letter.
 - (c) absence during which regular remuneration is paid.

"Fund" means the trust fund consisting of the property and

income therefrom which from time to time shall be received and held by the Trustee for the payment of benefits under the Plan.

"Highly-Compensated Sharing Member" means a Sharing Member who -----during the current Plan Year or the immediately preceding Plan Year:

- (a) was a five-percent owner, as defined in Section 416(i)(1) of the Code;
- (b) received more than \$75,000 (or such other amount as may be provided under Section 414(q) of the Code) in compensation from the Company or an Affiliate;
- (c) received more than \$50,000 (or such other amount as may be provided under section 414(q) of the Code) in compensation from the Company or an Affiliate and was among the

5 08/13/96

top 20% of Employees ranked by pay (excluding, for purposes of identifying the number of Employees in the top 20%, Employees described in Section 414(q)(8) of the Code to the extent (1) permitted under the Code and regulations thereunder and (2) elected by the Company); or

(d) was among the 50 (or, if lesser, the greater of 3 or 10% of all Employees, excluding, for purposes of identifying the number of Employees in the top 20%, Employees described in Section 414(q)(8) of the Code, to the extent (1) permitted under the Code and regulations thereunder and (2) elected by the Company) officers of the Company or an Affiliate who received compensation of more than 50% of the dollar limit under Section 415(b)(1)(A) of the Code; provided, however, that, if no officer has satisfied the compensation requirement described above during either the current Plan Year or the

immediately preceding Plan Year, the highest paid officer for such year shall be treated as a Highly-Compensated Sharing Member.

However, a Sharing Member, other than a five-percent owner, who was not a Highly-Compensated Sharing Member in the preceding year is a Highly-Compensated Sharing Member for the current year only if he is among the top 100 Employees ranked by compensation for the current year. For purposes of this Paragraph, "compensation" shall mean compensation reported on Form W-2 for the Employee for the Plan Year, but including amounts that would be excluded from an Employee's gross income

6 08/13/96

under a plan described in Section 125, 401(k), or 403(b) of the Code. Notwithstanding anything in this Plan to the contrary, Highly-Compensated Sharing Members shall be determined in accordance with Sections 414(q) and 401(k) of the Code and regulations thereunder.

"Hour of Service" means a credit used to measure service for ------various purposes under the Plan. Hours of Service are credited as follows:

- (a) Each week in which an Employee has any Full-Time Employment shall count as 45 Hours of Service.
- (b) Each week of absence for military service from which the Employee returns to the Company or an Affiliate with legally protected reemployment rights, but which is not treated as Full-time Employment, shall count as a number of Hours of Service equal to the number of hours of work in the Employee's customary week of work at the time the absence began.
- (c) Each hour which is not included in a period described in Paragraph (a) or (b) but for which an Employee is directly or indirectly paid or entitled to payment by the Company or an Affiliate, for the performance of duties or otherwise, including back pay, without regard to mitigation of damages, shall count as one Hour of Service; provided that no Hours of Service shall be credited under this Paragraph (c) to the extent such credit will cause the Employee to be credited with more than 501 Hours of Service (including Hours of Service credited under

08/13/96

Paragraphs (a) and (b)) with respect to any single continuous period during which the Employee performs no duties.

7

(d) Hours of Service for the performance of duties shall be credited to the Employee for the computation period or periods in which the services are performed. Hours of Service for back pay shall be credited to the Employee for the computation period or computation periods to which the award or agreement pertains rather than the computation period or periods in which it was

made. All other Hours of Service under Paragraph (c) shall be credited to the Employee for the computation period or periods in which payment is made or amounts payable to the Employee become due.

- (d) Paragraphs (a)-(d) notwithstanding, Hours of Service shall be credited at least as liberally as required by Department of Labor Regulation (S) 2530.200b-2(b) and (c).
- (f) In the case of an individual who is an Employee solely by reason of service as a leased employee, within the meaning of Section 414(n) or 414(o) of the Code, Hours of Service shall be credited as if such Employee were employed and paid with respect to such service (or with respect to any related absences or entitlements) by the Company or Affiliate that is the recipient thereof.

"Key Employee" means, with respect to any Plan Year, a key -----employee who is described in Section 416(i) of the Code.

8 08/13/96

"Member" means (a) an Active Member and (b) a former Active
---Member who has a balance standing to his Account under the Plan.

"Nonelective Contribution" means in any Top-Heavy Plan Year,

for each Active Member, a portion of the Company contribution allocated to the Active Member for the year under Section 4.1 with respect to which the deferral election under Section 4.2 shall not apply.

"Normal Retirement Age" means Age 65.

"Permissive Aggregation Group" means the group of qualified

retirement plans maintained by the Company or any Affiliate, which group consists of the Required Aggregation Group and any other such plan or plans which, considered together with the Required Aggregation Group, meets the requirements of Sections 401(a)(4) and 410 of the Code.

"Plan" means the Profit Sharing Plan set forth herein, as the $\hfill\Box$ same may be amended from time to time.

"Plan Year" means a calendar year after 1960.

"Predecessor Plan" means the Hourly Employees' Profit Sharing

Plan of P.H. Glatfelter Company and The Glatfelter Pulp Wood Company and any plan defined as a "Predecessor Plan" therein.

"Required Aggregation Group" means the group of qualified

retirement plans maintained by the Company or any Affiliate, including a frozen plan or a plan that has been terminated during the five year period ending on the last day of

9 08/13/96

the preceding Plan Year (or, in the case of the first Plan Year, the last day of such Plan Year), which group consists of each such plan in which a Key Employee is a participant (or, in the case of a terminated plan, was a participant in such five-year period) and each other plan that enables any such plan to meet the requirements of Section 401(a)(4) or 410 of the Code, but only if such group includes this Plan. Otherwise, the Required Aggregation Group consists of this Plan only.

"Separation from Service" means an Employee's resignation,

discharge or actual retirement from the Company, or any absence that causes him to cease to be an Employee.

"Sharing Member" means, for any Plan Year, an Employee who (a)

is an Active Member as of the last day of his Benefit Accrual Period for such Plan Year, and (b) completed 1,000 or more Hours of Service during the Plan Year or completed during his Benefit Accrual Period of the Plan Year a number of Hours of Service as an Active Member at least equal to 83-1/3 times the number of full calendar months in his Benefit Accrual Period.

"Super Top-Heavy Plan Year" means a Plan Year in which the Plan

is super top-heavy, determined by substituting "90%" for "60%" each place the latter percentage appears in the definition of Top-Heavy Plan Year.

"Top-Heavy Plan Year" means a Plan Year that begins after

December 31, 1983, in which the Plan is top-heavy. The Plan is top-heavy in any Plan Year if, on the last day of the

10 08/13/96

preceding year (or, in the case of the first Plan Year, the last day of such Plan Year), the Required Aggregation Group is top-heavy and the Required Aggregation Group is not part of a Permissive Aggregation Group that is not top-heavy. The Required Aggregation Group or a Permissive Aggregation Group (the "Group") is top-heavy for a given Plan Year if the aggregate of the accounts (or the present value of cumulative accrued benefits, in the case of a defined

benefit plan included in the Group) of participants who are Key Employees exceeds 60% of the like amount determined for all participants in all plans included in such Group. For purposes of this definition:

- (a) the account or the present value of the accrued benefit of any participant shall be increased by the amount of any distribution to such participant during the five-year period ending on the date described above; provided that no such increase shall arise from any rollover contribution or plan-to-plan transfer from this Plan that is not initiated by the participant or is made to another plan maintained by the Company or an Affiliate;
- (b) the account or the present value of the accrued benefit of a participant who has been a Key Employee but no longer is a Key Employee shall not be taken into account;
- (c) the account or present value of the accrued benefit of any participant who has not performed services for the Company or an Affiliate at any time during the five-year period

11 08/13/96

that ends on the determination date shall not be taken into account;

- (d) any rollover contribution from an unrelated plan, initiated by the Employee, and made after December 31, 1983 shall not be taken into account; and
- (e) effective January 1, 1987, the present value of accrued benefits under any defined benefit plan shall be determined under the method used for accrual purposes for all defined benefit plans maintained by the Company and all Affiliates if a single method is used by all such plans or, otherwise, the slowest accrual method permitted under Section 411(b)(1)(C) of the Code.

"Trust Agreement" means the agreement of trust between P.H.

Glatfelter Company and Morgan Guaranty Trust Company of New York, Trustee, prior to October 1, 1995, and the agreement of trust between P.H. Glatfelter Company and Fidelity Management Trust Company effective October 1, 1995, or any similar agreement hereafter made by P.H. Glatfelter Company and any Trustee at any time acting thereunder.

"Trustee" means Morgan Guaranty Trust Company of New York
----prior to October 1, 1995. Effective October 1, 1995, "Trustee" means Fidelity

Management Trust Company or its successor as Trustee under the Trust Agreement.

"Valuation Date" means December 31st and such other date or

2.4

12 08/13/96

1.2 Uncapitalized Terms. Any term used herein without an

initial capital letter that is used in a provision of the Code with which this Plan must comply to meet the requirements of Section 401(a) of the Code, and with which the Plan would not comply if such term were not interpreted as used in such provision, shall have as used herein the meaning that it has so used in such provision of the Code.

1.3 Gender. Whenever used herein, the masculine pronoun ----- shall include the feminine.

13 08/13/96

SECTION 2

ELIGIBILITY

- 2.1 Active Members as of December 31, 1988. Each Eligible Employee ______ who was an Active Member in the Plan as of December 31, 1988, shall be an Active Member as of January 1, 1989.

- determining whether an Employee has completed the service requirements of Section 2.2 or 2.3, employment prior to January 31, 1979, with Bergstrom Paper Company or any affiliate thereof (or a predecessor of either) shall be treated as employment with the Company; provided that only an Employee who is an Eliqible Employee shall become an Active Member.

Service with Bergstrom Paper Company. For purposes of

SECTION 3 _____

COMPANY CONTRIBUTIONS

P.H. Glatfelter Company. P.H. Glatfelter Company shall contribute for each Plan Year out of current or accumulated profits an amount determined as follows:

3.1

- (a) "Net Income" shall mean the unconsolidated earnings of P.H. Glatfelter Company as reported on for such Plan Year by the regularly employed certified public accountants of the Company in accordance with generally accepted accounting principles (1) before provision is made for Federal and State taxes based on income and any amounts paid or payable as profit sharing distributions, and (2) after such adjustment, if any, of such earnings as the Company may deem proper to exclude from income or costs and expenses on the internal financial statements of P.H. Glatfelter Company (A) any item or items attributable to operation of the Company's Ecusta or Bergstrom Paper Company facilities, for which purpose any interest charged against the Ecusta or Bergstrom Paper Company facilities for internal accounting purposes shall be treated as an item of expense attributable to those facilities but shall be recognized in earnings as an item of income not attributable to those facilities; and (B) the whole or any part of any item or items of an unusual or nonrecurring nature.
 - (b) The amount to be contributed shall equal the product of:

15 08/13/96

- (1) P.H. Glatfelter Company's Hourly Spring Grove Group percentage described in Section 3.3, multiplied by
 - (2)the sum of:
 - (A) 3% of the first \$2,000,000 of Net Income, plus
 - (B) 4% of the next \$1,000,000 of Net Income, plus
 - (C) 5% of the next \$1,000,000 of Net Income, plus
 - (D) 6% of the next \$1,000,000 of Net Income, plus
 - (E) 7% of all Net Income in excess of \$5,000,000;

provided that such sum for such Plan Year shall not exceed the lesser of (1) the

maximum amount which, after allowance for any contribution required by any pension or retirement plan of the Company, is an allowable deduction for Federal Income and Excess Profits Tax purposes, or (2) 15% of the aggregate Earnings received from the P.H. Glatfelter Company by all Sharing Members employed by such company during the Plan Year.

- (a) Compute the amount to be contributed by P.H. Glatfelter Company pursuant to Section 3.1.

16 08/13/96

- (b) Divide the amount to be contributed by P.H. Glatfelter Company by the aggregate Earnings received from such company by all Sharing Members employed by it during such Plan Year, thereby obtaining the contribution as a percent of the aggregate Earnings of all Sharing Members from such company.
- (c) The amount to be contributed by The Glatfelter Pulp Wood Company shall equal the percent obtained in (b) times the aggregate Earnings received from The Glatfelter Pulp Wood Company by all Sharing Members employed by it during such Plan Year; provided that the amount to be contributed for such Plan Year shall not exceed the lesser of (1) the maximum amount which, after allowance for any contribution required by any pension or retirement plan of the Company, is an allowable deduction for Federal Income and Excess Profits Tax purposes, or (2) 15% of the aggregate Earnings received from the Glatfelter Pulp Wood Company by all Sharing Members employed by it during the Plan Year.
 - 3.3 Hourly Spring Grove Group Percentage. The P.H. Glatfelter

Company's Hourly Spring Grove Group percentage for any Plan Year is the aggregate Earnings of all Sharing Members employed by P.H. Glatfelter Company for such Plan Year expressed as a percentage of the aggregate such Earnings of:

- (a) all Sharing Members in this Plan,
- (b) effective before October 1, 1995, all Sharing Members in the Profit Sharing Plan of P.H. Glatfelter Company and The Glatfelter Pulp Wood Company (Spring Grove Salaried Group),

17 08/13/96

or, effective October 1, 1995, all Profit Sharing Participants in the P.H. Glatfelter Company 401(k) Savings Plan who are members of the Spring Grove Salaried Group, and

(c) effective before October 1, 1995, all Sharing Members in

the P.H. Glatfelter Company Profit Sharing Plan (Corporate Group), or, effective October 1, 1995, all Profit Sharing Participants in the P.H. Glatfelter Company 401(k) Savings Plan, who hold positions in the Corporate Group associated with the Company's Spring Grove operations.

3.4 Nonforfeitability. Notwithstanding any other provision of the

Plan, each Member shall at all times have a 100% nonforfeitable interest in his Account Balance.

18 08/13/96

SECTION 4

ALLOCATIONS TO SHARING MEMBERS

4.1 Allocation.

- (a) As of December 31st of each Plan Year, the amount contributed by the Company in respect of such Plan Year shall be allocated to each Sharing Member for that Plan Year in that proportion which each such Sharing Member's Earnings for such Plan Year bears to the aggregate such Earnings of all Sharing Members for the Plan Year.
- (b) If for any Top-Heavy Plan Year the contribution on behalf of each individual who was an Active Member at any time during the Plan Year and who is not a Key Employee does not equal at least the required minimum percentage of compensation described in Subparagraph (1), below, the contribution under this Plan on account of each such Active Member (who is not an Employee subject to a collective bargaining agreement where retirement benefits were the subject of good faith bargaining) shall be increased by the amount necessary to cause such contribution to equal such minimum percentage. Contributions shall only be increased under the preceding sentence to the extent required under Section 416(c) of the Code, taking into account all defined contribution plans maintained by the Company. Notwithstanding any provision in the Plan to the contrary, in any Top-Heavy Plan Year the required minimum contribution determined under this Paragraph shall be a Nonelective Contribution for the year, and shall be allocated to

19 08/13/96

a separate Account for each such Active Member. For purposes of this Section:

(1) the required minimum percentage shall be 3%; provided, that, if the Plan does not enable a defined benefit plan in the Required Aggregation Group to meet the requirements of Section 401(a)(4) or 410 of the Code, the required minimum percentage shall be the lesser of 3% or the

percentage of compensation, to a maximum of \$200,000 (for Plan Years beginning prior to January 1, 1994) or \$150,000 (for Plan Years beginning on or after January 1, 1994) of compensation (or such other amount as may be fixed by the Secretary of the Treasury or his delegate pursuant to Section 401(a)(17) of the Code), allocated to the Key Employee for whom such percentage is the highest for such Plan Year;

- (2) "compensation" shall mean compensation reported on Form W-2 for the Employee for the Plan Year;
- (3) contributions under any defined contribution plan included in the Required Aggregation Group or any Permissive Aggregation Group shall be treated as contributions under this Plan; provided, however, that elective deferrals made on behalf of individuals who are not Key Employees shall not be used to meet the minimum contribution requirements of this Paragraph (c); and
- (4) If an Active Member who is not a Key Employee participates in both a defined benefit plan and a defined contribution plan in the Required Aggregation Group, the

20 08/13/96

Active Member shall receive the minimum benefit provided under the defined benefit plan in lieu of the minimum contribution provided hereunder.

- 4.2 Deferred Elections. Subject to Section 4.1(b) and Section
- 6.2(b)(3), and according to procedures adopted by the Administrator, for each Plan Year each Sharing Member shall be entitled to elect to have any whole percentage, of between 1% and 100%, inclusive, of his Elective Contribution for that year paid over to the Trustee for credit to his Account. The amount so deferred shall be his "elective deferral" for that Plan Year unless an adjustment is required by the limitations of Section 4.3. Any such election shall be filed in a form acceptable to the Administrator according to procedures adopted by the Administrator and an election once filed shall be binding on and irrevocable by the Sharing Member for the Plan Year to which it relates. election filed with the Administrator shall be presumed to continue in force from year to year until a revised election is filed. The Administrator shall notify each Member, prior to December 15 of each year, of his right to file a revised election. In the absence of an effective election a Sharing Member's elective deferral shall be 100% of his Elective Contribution.
- (a) For any Plan Year beginning after December 31, 1992, the average actual deferral percentage for the

Highly-Compensated Sharing Members who are covered under this Plan pursuant to a collective bargaining agreement shall not exceed the greater of:

(1) one hundred twenty-five percent (125%) of the average actual deferral percentage for all other Sharing Members who are covered under this Plan pursuant to a collective bargaining agreement; or

(2) the lesser of:

- (A) two hundred percent (200%) of the average actual deferral percentage for all other Sharing Members who are covered under this Plan pursuant to a collective bargaining agreement; or
- (B) two percent (2%) plus the average actual deferral percentage for all other Sharing Members who are covered under this Plan pursuant to a collective bargaining agreement.
- (b) For any Plan Year beginning after December 31, 1986, the average actual deferral percentage for the Highly-Compensated Sharing Members who are covered under this Plan without regard to a collective bargaining agreement shall not exceed the greater of:
- (1) one hundred twenty-five percent (125%) of the average actual deferral percentage for all other Sharing Members who are covered under this Plan without regard to a collective bargaining agreement; or
 - (2) the lesser of:

22 08/13/96

- (A) two hundred percent (200%) of the average actual deferral percentage for all other Sharing Members who are covered under this Plan without regard to a collective bargaining agreement; or
- (B) two percent (2%) plus the average actual deferral percentage for all other Sharing Members who are covered under this Plan without regard to a collective bargaining agreement.
- (c) For purposes of Paragraph (a), the "average actual deferral percentage" for a specified group of Sharing Members for a given Plan Year shall be the average of the actual deferral percentages, calculated separately for each Sharing Member in such group. The "actual deferral percentage" shall be the ratio of:
- (1) the sum of the amount of Elective Contributions actually paid over to the Fund on behalf of each Sharing Member for such Plan Year (including such amounts as may be paid to the Sharing Member under

Paragraph (f) for the Plan Year), plus the amount of Nonelective Contributions allocated to each Sharing Member for such Plan Year (if the Company elects to take such Nonelective Contributions into account) and, in the case of any Sharing Member who is a Highly-Compensated Sharing Member, elective deferrals for the year under any other qualified retirement plan maintained by the Company or any Affiliate; to

(2) the Sharing Member's compensation (as defined in Paragraph (e)) for such Plan Year.

23 08/13/96

- (d) (1) In any Plan Year in which the elections by Members under Section 4.2 would otherwise cause both of the tests described in Paragraph (a) or (b) to be failed, the elective deferrals by Highly-Compensated Sharing Members shall be reduced to the extent necessary to satisfy at least one of the tests in Paragraph (a). The reduction shall be accomplished (i) by prospectively limiting the amount of elective deferrals by Highly-Compensated Sharing Members or, (ii) after application of clause (i) of this sentence, by paying to the Highly-Compensated Sharing Members the excess amount of their elective deferrals (with earnings thereon) not later than the end of the Plan Year immediately following the close of the Plan Year in which the elective deferrals were made; provided, however, that, for any Highly-Compensated Sharing Member who is also a participant in any other qualified retirement plan maintained by the Company or any Affiliate under which the Highly-Compensated Sharing Member makes elective deferrals for such year, the Company shall coordinate corrective action under this Plan and such other plan for the year.
- (2) The amount of the reduction shall be the amount necessary to reduce the amount of elective deferrals by Highly-Compensated Sharing Members to a maximum adjusted percentage, which shall be the highest percentage that would cause one of the tests in Paragraph (a) or Paragraph (b), as the case may be, to be satisfied if each such Highly-Compensated Sharing Member who had an actual deferral percentage greater than

24 08/13/96

such maximum adjusted percentage had instead such lower percentage, reducing the elective deferrals of the Highly-Compensated Sharing Members and elective deferrals under any other qualified retirement plan maintained by the Company or an Affiliate (less any amount previously distributed under Paragraph (g) for the year), in order, beginning with the Highly-Compensated Sharing Member with the highest actual deferral percentage.

(e) For any Plan Year in which this Plan and any other qualified retirement plan of the Company or an Affiliate which includes a cash or deferred arrangement are aggregated in order to satisfy the requirements of Sections 401(a)(4), 410(b) or 401(k) of the Code, the identification of Sharing Members and Highly-Compensated Sharing Members for the year shall be determined on the basis of all Employees eligible to make an election under the cash or

deferred arrangements of all such plans and the provisions of this Section 4.3 shall be applied as if all such plans were one plan.

- (f) For purposes of this Section:
- (1) "compensation" shall mean compensation reported on Form W-2 for the Employee for the Plan Year, and, at the election of the Company, may include any amount contributed by the Company or an Affiliate on behalf of Sharing Members as elective deferrals or pursuant to salary reduction agreements that is not includible in the gross income of Sharing Members under Section 125 or 402(a)(8) of the Code;

25 08/13/96

- (2) with respect to any Plan Year, only compensation not in excess of the amount to which the \$200,000 (for Plan Years beginning before January 1, 1994) or the \$150,000 (for Plan Years beginning on or after January 1, 1994) limit of Code section 401(a) (17) has been indexed shall be taken into account. In determining compensation for purposes of this limitation, the rules of section 414(q) (6) of the Code shall apply, except in applying such rules, the term "family" shall include only the spouse of the Employee and any lineal descendants who have not attained Age 19 before the close of the Plan Year. In applying the rules of section 414(q) (6) of the Code, the limit of this Clause shall be allocated among family members in proportion to their compensation as defined in Clause (1); and
- (3) the compensation and elective deferrals of any 5% owner or any other Highly-Compensated Sharing Member who is one of the top 10 Employees ranked by pay for the year shall be increased by the amount of the compensation or elective deferrals, whichever is applicable, of any Employee who is a spouse or lineal ascendant or descendant (or a spouse thereof) of such Highly-Compensated Sharing Member, and such family member shall not be considered a separate Employee.
- (g) Effective for Plan Years beginning on or after January 1, 1988, elective deferrals under this Plan, plus any other elective deferrals (as defined in Section 402(g)(3) of the Code) made under any other plan or arrangement maintained by

26 08/13/96

the Company or an Affiliate for a Sharing Member shall not exceed \$7,000 (or such other amount as may apply under Section 402(g) of the Code) for any calendar year. To the extent necessary to satisfy this limitation for any year (1) elections under Section 4.2 shall be prospectively restricted, and (2) after application of clause (1), the excess elective deferrals under this Plan and any other plan maintained by the Company or an Affiliate (with earnings thereon, but reduced by any amounts previously distributed under Section 4.3(c) for the year) shall be paid to the Member on or before the April 15 first following the

calendar year in which such elective deferrals were made. If the elective deferrals under this Plan and any other plan or arrangement maintained by the Company or an Affiliate plus the elective deferrals under any other plan for a Sharing Member exceed such limitation for any calendar year, upon the written request of the Sharing Member made on or before the March 1 first following such calendar year, the excess, including any earnings attributable thereto, shall be paid to the Sharing Member on or before the April 15 first following such calendar year.

- (h) The application of this Section 4.3 shall satisfy Section 401(k) of the Code and regulations thereunder and such other requirements as may be prescribed by the Secretary of the Treasury.

27 08/13/96

each Plan Year for which contributions are made pursuant to Sections 3.1 and 3.2, the Nonelective Contribution, if any, of each Active Member for such Plan Year shall be paid over to the Trustee and paid to the Member's Account. The balance of the Sharing Member's Elective Contribution over his elective deferral for the Plan Year shall be paid to him in cash (net of any taxes withheld thereon) not later than the March 1 next following the close of the Plan Year for which the allocation was made. The elective deferral of each Sharing Member for the Plan Year shall be paid over to the Trustee as soon as practicable, but in no event later than 90 days after the date the Sharing Member would have received such amount in cash if he had so elected.

from allocations for prior Plan Years shall have been distributed pursuant to Paragraph 7.1(a), then amount, otherwise payable to the Trustee pursuant to Section 4.4 shall be paid in cash directly to such Member, or, if deceased, to his Beneficiary.

Cash Payments. If the entire Account of any Member resulting

4.6 Maximum Allocations.

4.5

- (a) Effective for Plan Years that begin after January 15, 1984 and before January 19, 1987, allocations of Company contributions to Members' Account shall be made subject to the following rules.
- (1) During any 12-month period beginning May 1 and ending April 30 ("Limitation Year"), the Company

08/13/96

contributions allocated to a Member's Account ("Annual Addition") shall not exceed the least of:

(A) \$30,000 (or such other maximum amount as may be fixed for qualified defined contribution plans by the Secretary of the Treasury or his delegate); or

(B) 25% of his compensation from the Company and any Affiliate for the Year; or

(C) his Combined Plan Maximum as defined in Subparagraph (a)(2).

The Annual Addition to a Participant's Account is the sum, for all qualified defined contribution plans of the Company or any Affiliate, of (i) employer contributions and forfeitures allocated to his Account for the Limitation Year; plus (ii) the lesser of 50% of his own contributions for the Limitation Year or the amount by which such contributions exceed 6% of his total compensation from the Company or an Affiliate; plus (iii) any contribution allocated to an individual medical account (as defined in Section 415(1)(2) of the Code) under a defined benefit plan maintained by the Company or an Affiliate; plus (iv) amounts derived from contributions paid or accrued after December 31, 1985, which are attributable to post-retirement medical benefits allocated to the separate account of a Key Employee under a welfare benefit fund, as defined in Section 419(e) of the Code, maintained by the Company or an Affiliate.

(2) A Member's Combined Plan Maximum for a Limitation Year means the amount which causes his Defined

29 08/13/96

Contribution Plan Fraction for the year to be the difference between 1.0 and his Defined Benefit Plan Fraction for the year. The terms in the preceding sentence have the following meaning:

(A) The "Defined Benefit Plan Fraction" is a fraction the numerator of which is the Member's projected annual benefit attributable to employer contributions under any qualified defined benefit plan to which the Company or an Affiliate contributes or has contributed on his behalf and the denominator of which is the lesser of 1.25 multiplied by \$90,000 (or such other dollar limitation in effect for the Limitation Year under Section 415(b)(1)(A) of the Code) adjusted to reflect commencement of benefits before Age 62 or after Age 65, or 1.4 multiplied by 100% of his average annual compensation from the Company or any Affiliate in the thirty-six consecutive months which yield the highest average.

(B) The percentage limitation for any Limitation Year shall be 25% of the Member's compensation for that Limitation Year from the Company or any Affiliate (or such other percentage as may have been applied

under such plans as in effect on the last day of any prior year or may be prescribed by the Secretary of the Treasury or his delegate for any future year).

(3) In any Top-Heavy Plan Year and in any Super Top-Heavy Plan Year, Subparagraph (a)(2) above shall be applied by substituting "1.0" for "1.25" in the definitions of Defined Benefit Plan Fraction and Defined Contribution Plan

30 08/13/96

Fraction therein, unless in any Top-Heavy Plan Year (but not in any Super Top-Heavy Plan Year) Section 4.1(a) would not require an increase in the Company contribution allocated to any Member if it were applied by substituting "4%" for "3%" therein.

- (4) For purposes of this Section 4.6(a), "compensation" shall mean the participant's wages, salaries, fees for professional service and other amounts received for personal services actually rendered in the course of employment with the Company or any Affiliate, but excluding all other forms of remuneration.
- (b) Effective for Plan Years that begin after January 19, 1987, allocations of Company contributions to Members' Account shall be made subject to the following rules.
- (1) During the period beginning May 1, 1987 and ending December 31, 1987 and each calendar year thereafter ("Limitation Year"), the Company contributions allocated to a Member's account under this Plan and any other defined contribution plan maintained by the Company or an Affiliate ("Annual Addition") shall not exceed the least of:
- (A) the greater of (i) \$30,000 or (ii) one-quarter of the maximum dollar amount fixed for qualified defined benefit plans under Section 415(b)(1)(A) of the Code by the Secretary of the Treasury or his delegate; provided, however, that, for any Limitation Year that is a period of less than 12 months, the amount otherwise applicable under this Subparagraph (1)(A) shall be multiplied by a fraction, the numerator of which

31 08/13/96

is the number of months in the Limitation Year and the denominator of which is 12; or

- (B) 25% of the compensation reported on Form W-2 for the Employee for the Limitation Year; or
- (C) his Combined Plan Maximum as defined in Subparagraph (b) (2).

The Annual Addition to a Member's Account is the sum, for all qualified defined contribution plans of the Company or any Affiliate, of (i) employer contributions and forfeitures allocated to his Account for the Limitation Year; plus (ii) his own contributions for the Limitation Year; plus (iii) any contribution allocated to an individual medical account (as defined in Section 415(1)(2) of the Code) under a defined benefit plan maintained by the Company or an Affiliate; plus (iv) amounts derived from contributions paid or accrued after December 31, 1985, which are attributable to post-retirement medical benefits allocated to the separate account of a Key Employee under a welfare benefit fund, as defined in Section 419(e) of the Code, maintained by the Company or an Affiliate.

(2) A Member's Combined Plan Maximum for a Limitation Year means the amount which causes his Defined Contribution Plan Fraction for the year to be the difference between 1.0 and his Defined Benefit Plan Fraction for the year. The terms in the preceding sentence have the following meaning:

(A) The "Defined Benefit Plan Fraction" is a fraction the numerator of which is the Member's projected

32 08/13/96

annual benefit attributable to employer contributions under any qualified defined benefit plan to which the Company or an Affiliate contributes or has contributed on his behalf and the denominator of which is the lesser of 1.25 multiplied by \$90,000 (or such other dollar limitation in effect for the Limitation Year under Section 415(b)(1)(A) of the Code) adjusted to reflect commencement of benefits before or after the Member's Social Security Retirement Age (as defined below), or 1.4 multiplied by 100% of his average annual compensation (as defined in Section 415(c)(3) of the Code) from the Company or any Affiliate in the thirty-six consecutive months which yield the highest average. For purposes of this Section, a Member's Social Security Retirement Age is (i) Age 65, if the Member was born before January 1, 1938, (ii) Age 66, if the Member was born after December 31, 1937, but before January 1, 1955, or (iii) Age 67, if the Member was born after December 31, 1954.

(B) The "Defined Contribution Plan Fraction" is a fraction the numerator of which is the sum of the Annual Additions to the Member's account under all qualified defined contribution plans to which the Company or an Affiliate contributes or has contributed on his behalf for the Limitation Year and all prior Limitation Years, and the denominator of which is the maximum aggregate amounts for all Limitation Years in which he has been an Employee. A Member's maximum aggregate amount for any Limitation Year shall equal the lesser of 1.25 multiplied by the dollar limitation for such Limitation Year or

33 08/13/96

1.4 multiplied by the percentage limitation for such Limitation Year. The dollar limitation for any Limitation Year is \$30,000 (or such other amount as may have

been applied under such plans as in effect on the last day of any prior year or, for any future year, if greater, one-quarter of the dollar limitation that may be prescribed by the Secretary of the Treasury or his delegate under Section 415(b)(1)(A) of the Code); provided, however, that, for any Limitation Year that is a period of less than 12 months, the dollar limitation that is otherwise applicable shall be multiplied by a fraction, the numerator of which is the number of months in the Limitation Year and the denominator of which is 12. The percentage limitation for any Limitation Year shall be 25% of the Member's compensation (as defined in Section 415(c)(3) of the Code) for that Limitation Year from the Company or any Affiliate (or such other percentage as may have been applied under such plans as in effect on the last day of any prior year or may be prescribed by the Secretary of the Treasury or his delegate for any future year).

(C) In any Top-Heavy Plan Year and in any Super Top-Heavy Plan Year, Subparagraph (b)(2) above shall be applied by substituting "1.0" for "1.25" in the definitions of Defined Benefit Plan Fraction and Defined Contribution Plan Fraction therein, unless in any Top-Heavy Plan Year (but not in any Super Top-Heavy Plan Year) Section 4.1(a) would not require an increase in the Company contribution allocated to any Member if it were applied by substituting "4%" for "3%" therein.

34 08/13/96

- (c) The rules of this Section 4.6 shall be applied in accordance with Section 415 of the Code and regulations thereunder.
- (d) Any amount which would be allocated to a Sharing Member, but for the limitations of this Section 4.6, shall be paid to the Sharing Member in cash. However, if the amounts allocated to the account of a Member under this Plan and any other defined contribution plan maintained by the Company or an Affiliate exceed the limitation set forth in this Section as a result of a reasonable error in estimating the Member's compensation, or such other circumstances as may be permitted by law, the Company shall determine which portion, if any, of such excess amount is attributable to a Member's elective deferrals and Nonelective Contributions under the Plan, until such amount has been exhausted, and shall take the following appropriate steps to correct such violation:
- (1) While the Member remains an Employee, his excess Nonelective Contributions and/or elective deferrals shall be held in a suspense account by the Trustee until the following Plan Year (or any succeeding Plan Years), at which time such amounts shall be allocated to the Member's Account before any Nonelective Contributions or elective deferrals, as appropriate, are made on his behalf for such Plan Year; and
- (2) When the Member ceases to be an Employee, his excess Nonelective Contributions and/or elective deferrals, along with earnings thereon, held in the suspense

account shall be allocated in the following Plan Year (or any succeeding Plan Years) to the Accounts of all Sharing Members before any contributions by the Company or Sharing Members may be made for such Plan Year.

35

All amounts held in a suspense account pursuant to this Paragraph (d) shall share in investment gains and losses of the Fund.

4.7 Predecessor Plan. As to Members of a Predecessor Plan who

become Members of this Plan, this Plan is a continuation of such Predecessor Plan. All Account Balances held as of December 31, 1981, under the Predecessor Plan for Members of the Predecessor Plan who become Members of this Plan shall be deemed to be held under this Plan as part of each such Member's Account.

36 08/13/96

SECTION 5

ADMINISTRATION OF THE FUND

5.1 Investment by Trustee. All Company contributions which are

paid over to the Trustee shall be invested by the Trustee upon such terms as may be set forth in the Trust Agreement. The Fund under this Plan may be commingled for investment purposes with the fund under any other qualified profit sharing plan to which P.H. Glatfelter Company contributes.

5.2 Investment Portfolios. The Administrator shall, by non-

discriminatory rule, divide the Fund into two or more separate investment portfolios, and authorize Members to direct how their Accounts shall be invested among the portfolios.

5.3 Valuation of Accounts. The value of each investment portfolio in the Fund shall be computed by the Trustee as of the close of business on each Valuation Date on the basis of the fair market value of the assets of the Fund. The value of each investment portfolio in the Fund shall be compared with the value of such investment portfolio in the Fund as of the preceding Valuation Date. Any difference in the value, not including contributions, distributions, fees, and transfers made since the preceding Valuation Date, shall be the net increase or decrease of such portfolio in the Fund, and such amount shall be ratably apportioned by the Trustee on its books, among the Members' Accounts which are invested in such investment portfolio at the current Valuation Date.

BENEFITS FOR MEMBERS

6.1 Separation from Service or Death. A Member's Account Balance

shall be distributed as soon as administratively practicable after his Separation from Service or death; provided, however, that, except in the case of a distribution due to the Member's death, if a Member's Account Balance as of the proposed date of distribution exceeds, or has exceeded at the time of any prior distribution, \$3,500, benefit commencement shall be deferred until the Member attains Normal Retirement Age unless the Member elects, according to procedures adopted by the Administrator, an earlier benefit commencement date. Such election may be made no earlier than 90 days prior to the benefit commencement date elected by the member, except as described below. The Company shall supply to each Member whose Account Balance exceeds, or has exceeded at the time of any prior distribution, \$3,500, written information relating to (1) his right to defer distribution; (2) the material features of the forms of payment available to him; and (3) the relative values of such forms of payment. Such notice shall be furnished not less than 30 days and not more than 90 days prior to the date of any distribution, including a distribution pursuant to Section 6.2 or 6.3, that occurs prior to the earlier of the Member's death or the date he attains Normal Retirement Age, except that such notice may be furnished and the distribution may be made less than 30 days prior to the date of distribution if (1) the

38 08/13/96

Administrator informs the Member that the Member has the right to a period of at least 30 days after receiving such notice to consider the decision whether to elect a distribution and the mode in which he desires such distribution to be made, and (2) the Member, after receiving such notice, affirmatively elects a distribution. The value of a Member's Account Balance for distribution purposes shall be determined at the Valuation Date as of which the distribution is made.

6.2 Emergency Withdrawals.

(a) Prior to April 1, 1989, if a Member establishes to the satisfaction of the Administrator his need for funds because of an emergency resulting from a personal accident or illness; an illness, accident or death within his immediate family; or other bona fide emergencies, he shall be entitled to a single-sum emergency benefit in an amount necessary to meet the emergency; provided that the emergency imposes an immediate and heavy financial need upon the Member, which need cannot be met from other resources reasonably available to the Member. The amount which may be withdrawn under this Section 6.2(a) may not exceed the net value of the Member's Account Balance less amounts attributable to (1) earnings on post-1979 elective deferrals credited to the

Member's Account after December 31, 1988 and (2) Nonelective Contributions which have been taken into account under Section 4.3(b)(1). The amount distributed will be deducted from and not exceed the net value of the portion of the Member's Account Balance which is eligible for withdrawal under this

39 08/13/96

Section 6.2(a) at the Valuation Date as of which the distribution is made.

- (b) (1) On or after April 1, 1989, if a Member establishes to the satisfaction of the Administrator his need for funds due to an emergency resulting from:
 - (A) (i) Unreimbursed medical expenses, or
- (ii) Income loss due to the inability or decreased ability of the Member or his spouse to perform his or her job, as a result of an accident or illness of the Member, his spouse or dependent;
- (B) Loss of income due to death of the Member's spouse;
- (C) Catastrophic damage to real or personal property of the Member where uninsured damage exceeds \$5,000; and
- (D) Such other circumstances or events in the nature of an unanticipated emergency as may be prescribed by the Secretary of the Treasury or his delegate, he shall be entitled to a single-sum emergency benefit, in an amount not exceeding the amount necessary to meet the emergency; provided that the emergency imposes an immediate and heavy financial need upon the Member, which need cannot be met from other resources reasonably available to the Member. The amount which may be withdrawn as an emergency benefit under this Section

40 08/13/96

6.2(b) may not exceed the net value of the Member's Account Balance less amounts attributable to (1) earnings on post-1979 elective deferrals credited to the Member's Account after December 31, 1988 and (2) Nonelective Contributions which have been taken into account under Section 4.3(b)(1). Amounts withdrawn shall be charged first against the portion of the Member's Account Balance that is attributable to the Member's (1) pre-1980 elective deferrals and (2) Nonelective Contributions which have not been taken into account under Section 4.3(b)(1), and then to the remainder of the Member's Account Balance which is eligible for withdrawal under this Section 6.2(b); provided that the amount distributed shall be deducted from and shall not exceed the value of the portion of the Member's Account Balance which is eligible for withdrawal under this Section 6.2(b) at the Valuation Date as of which the distribution is made.

(2) Notwithstanding any provision in the Plan to the

contrary, on or after April 1, 1989, a Member shall not be permitted to withdraw any amount attributable to post-1979 elective deferrals unless the Member has obtained all currently permissible distributions (other than distributions of elective deferrals for emergency or financial necessity) and non-taxable loans, if any, under this Plan and all other plans maintained by the Company or any Affiliate and the Member agrees to be bound by the rules set forth in Subparagraph (3).

(3) If the Member withdraws any amounts attributable to post-1979 elective deferrals under this Plan or

41 08/13/96

any other qualified retirement plan maintained by the Company or an Affiliate, which other plan conditions such withdrawal upon the Member's being subject to rules similar to those stated in this Section 6.2(b), such Member:

- (A) May not make elective deferrals under Section 4.2 of this Plan or employee contributions (other than mandatory employee contributions under a defined benefit plan) or elective deferrals under any other plan maintained by the Company or an Affiliate, including the P.H. Glatfelter Company Employee Stock Purchase Plan, for a period of twelve months commencing on the date of his receipt of the withdrawal; and
- (B) In the calendar year next following the calendar year of such withdrawal, may not make elective deferrals under this Plan or any other qualified retirement plan maintained by the Company or an Affiliate in excess of:
- (i) \$7,000, or such other amount as may be prescribed under Section 402(g) of the Code, minus
- (ii) The total elective deferrals made by the Member under this Plan and any other qualified plan during the calendar year of the withdrawal.
- 6.3 Other Withdrawals. A Member may make withdrawals from his account subject to the following rules:
- (a) During his fifth consecutive Plan Year as a Member and on or before December 31 thereof or according to other

42 08/13/96

administrative rules adopted by the Company, he may elect to withdraw in a single sum either 50% or 100% of his Account Balance, excluding the portion thereof attributable to post-1979 elective deferrals (i.e., the portion attributable to contributions to this Plan or a Predecessor Plan for Plan Years after 1979 that the Member could have elected under Section 4.2 to receive in cash) or his Nonelective Contributions. In addition, during each tenth

consecutive Plan Year thereafter he may again elect to withdraw in a single sum either 50% or 100% of his Account, excluding the portion thereof attributable to post-1979 elective deferrals or his Nonelective Contributions.

(b) An election authorized under Paragraph (a) shall be made in a form acceptable to the Administrator on or before December 31 of the Plan Year during which authorized; provided, however, that, if the Member's Account Balance exceeds, or has exceeded at the time of any prior distribution, \$3,500, such election may not be made earlier than 90 days prior to the date of withdrawal and any election so filed shall be irrevocable; except that distribution may be paid within 30 days following the election if (1) the Administrator informs the Member that the Member has the right to a period of at least 30 days after receiving such notice to consider the decision whether to elect a distribution and the mode in which he desires such distribution to be made, and (2) the Member, after receiving such notice, affirmatively elects a distribution. If he fails to make a timely election during a Plan Year in which authorized, all

43 08/13/96

amounts in his Account shall remain in the Fund to the credit of his Account and subject to withdrawal only as otherwise authorized by this Section 6.3.

- (c) Payment of the amounts withdrawn shall be made in cash and shall be based on the value of his Account as of the Valuation Date coincident with or immediately preceding the distribution.
- (d) (1) In addition to the foregoing, the Member may at any time after his fifth consecutive Plan Year as a Member request to withdraw from his Account in a single sum such amount as is necessary to meet a financial necessity; provided that the amount which may be withdrawn may not exceed the net value of the Member's Account Balance less amounts attributable to (1) earnings on post-1979 elective deferrals credited to a Member's Account after December 31, 1988 and (2) Nonelective Contributions which have not been taken into account under Section 4.3(b)(1).
 - (2) For purposes of this Paragraph:
- (A) Prior to April 1, 1989, a "financial necessity" shall mean the need for funds to meet the costs of higher education for the Member or a dependent of the Member, the purchase of a home for the Member and his family or similar substantial and nonrecurring expenses for the welfare of the Member and his family; and
- (B) On or after April 1, 1989, a "financial necessity" shall mean the need for funds:

44 08/13/96

(i) For costs directly related to the

purchase (excluding mortgage payments of) the Member's principal residence;

(ii) To prevent eviction from or the foreclosure on the mortgage on the Member's principal residence;

(iii) To pay tuition, related educational fees and room and board expenses for the next 12 months of post-secondary education for the Member, his spouse, or dependents (as defined in section 152 of the Code);

(iv) In connection with such other circumstances or events of a non-emergency nature as may be prescribed by the Secretary of the Treasury or his delegate;

(v) To complete major home maintenance or improvements costing \$1,000 or more;

(vi) To pay debts of \$1,000 or more which are substantially in excess of the Member's assets or income;

(vii) To pay expenses of \$1,000 or more related to divorce and other legal settlements; and

(viii) To pay funeral expenses of \$1,000 or more for a member of the Member's immediate family; provided that the financial necessity imposes an immediate and heavy financial need upon the Member which cannot reasonably be met from other resources available to the Member.

45 08/13/96

(3) The amount distributed will be deducted from and not exceed the net value of the portion of the Member's Account Balance which is eligible for withdrawal under this Section 6.3(d) at the Valuation Date as of which the withdrawal is made; provided that, on or after April 1, 1989, amounts withdrawn under this Section shall be charged first against the portion of the Member's Account Balance that is attributable to his (1) pre-1980 elective deferrals and (2) Nonelective Contributions which have not been taken into account under Section 4.3(b)(1), and then to the remainder of the Member's Account Balance which is eligible for withdrawal under this Section.

(4) Notwithstanding any provision in the Plan to the contrary, on or after April 1, 1989, a Member shall not be permitted to withdraw any amount attributable to his post-1979 elective deferrals unless the Member's financial necessity is due to circumstances described in Section 6.3(d)(2)(B)(i) through (v), and the requirements of Sections 6.2(b)(2) and (3) are met with respect to the Member.

METHOD OF DISTRIBUTION OF BENEFITS

7.1 Form of Distribution. All benefits provided by this Plan with

respect to a Member or former Member shall be distributed from such Member's Account commencing not later than sixty days following the close of the Plan Year in which his Separation from Service (for reasons other than the death of the Employee) occurs or, if distribution is deferred under Section 6.1 until attainment of Normal Retirement Age, actual distribution shall commence not later than sixty days following the close of the Plan Year in which the Member attains Normal Retirement Age. A Member may elect the method of distribution, subject to the following rules:

- (a) A Member's benefit may be paid to him in a single sum as soon as administratively practicable after his Separation from Service or, if distribution is deferred to a later date pursuant to Section 6.1, as of such later benefit commencement date elected by the Member.
- (b) Payments may be made in approximately equal monthly, quarterly, or annual installments over a period of time not to exceed the life expectancy of the Member or the joint and last survivor life expectancy of the Member and his designated Beneficiary, or such lesser period of time as may be required pursuant to the minimum distribution incidental benefit requirements of Section 401(a)(9)(G) of the Code and regulations thereunder.

- (c) Payments may not be made in the form of an annuity for the life of the Member or for any period measured with respect thereto.
- (d) Notwithstanding any other provision of the Plan, the entire Account Balance of a Member shall be distributed to him:
- (1) not later than April 1 of the calendar year immediately following the calendar year in which he attains Age 70-1/2 (his "Required Beginning Date"); or
- (2) commencing not later than his Required Beginning Date and over a period certain not extending beyond the joint and last survivor life expectancy of the Member and a designated Beneficiary, or such lesser period of time as may be required pursuant to the minimum distribution incidental benefit requirements of Section 401(a)(9)(G) of the Code and regulations thereunder.
 - 7.2 Death Benefits.
 - (a) Designation of Beneficiary. Each Member shall have a

Beneficiary who shall receive the death benefit, if any, payable under Paragraph (b) or (c). Each Member's Beneficiary shall be his surviving spouse unless:

(1) the Member has designated a different Beneficiary and the Member's spouse has consented to that designation in writing in an instrument that acknowledges the effect of the designation and is witnessed by a representative of the Administrator on behalf of the Plan or by a notary public; or

48 08/13/96

(2) the Administrator concludes that such consent cannot be obtained because the Member has no spouse or because the spouse cannot be located or that his spouse's consent is not required under such circumstances as are prescribed under governmental regulations.

If the Member has no spouse on the date of his death, his Beneficiary shall be the person or persons designated by the Member as contingent Beneficiary or Beneficiaries in a designation that named the Member's spouse as primary Beneficiary, or such other person or persons as the Member may have designated as primary or contingent Beneficiary in a designation made at a time when the Member had no spouse or which met the spousal consent requirements described above at the time such designation was made. If no individual named as Beneficiary survives the Member, and no other Beneficiary has been effectively designated by the Member, his Beneficiary shall be his estate.

A Member may change or revoke any designation made by him under this Section. Any such action, however, (including amendment of any instrument governing a trust designated as beneficiary) that results in someone other than his surviving spouse being his Beneficiary shall be deemed to have named such spouse as primary Beneficiary and such other person as secondary Beneficiary unless the requirements of clause (1) or (2) of this Paragraph are met with respect to such action. The requirements of clause (1) of this Paragraph shall be satisfied with respect

49 08/13/96

to a change of Beneficiary if the Member's spouse previously consented to the designation of an alternative Beneficiary in accordance with clause (1) and the spouse's consent acknowledged her right to limit consent to a specific Beneficiary and expressly permitted the Member to change his designation without further consent by the spouse.

(b) Death Before Required Beginning Date. If a Member dies

before his Required Beginning Date described in Section 7.1(d), his entire Account Balance (if there remains any amount in his Account) shall be paid to his Beneficiary. Payment shall be made in a lump sum, or in such other form permitted under Section 7.1 as the Beneficiary may elect; provided, that

payments shall be completed no later than the end of the calendar year in which the fifth anniversary of the Member's death occurs unless the Beneficiary is an individual, in which case distribution may be made in substantially equal installments that begin no later than December 31 of the calendar year following the calendar year in which the Member's death occurs (or, if the Beneficiary is the Member's spouse the later of (1) December 31 of the calendar year following the calendar year in which the Member's death occurs or (2) December 31 of the calendar year in which the Member would have attained Age 70-1/2) and continue over a period no longer than the life expectancy of the Beneficiary. If the Beneficiary is the Member's spouse, but such spouse dies before payments to the Beneficiary begin, payments to

50 08/13/96

a contingent Beneficiary may begin and shall end no later than the dates determined as if the spouse were the Member.

(c) Death After Required Beginning Date. If a Member dies

after his Required Beginning Date described in Section 7.1(d), any amount remaining in his Account shall be paid to his Beneficiary. Payment shall be made under the form of payment elected by the Member before payments began or in such other form permitted under Section 7.1 as the Beneficiary may elect, but at a rate no less rapid than under the form of payment elected by the Member.

expectancies shall be determined when installment payments begin and shall be redetermined each year thereafter in which such a redetermination would increase the payment period (so measured) by at least one year.

7.4 Minimum Distribution Requirements. Notwithstanding any

provision in the Plan to the contrary, the form and timing of any distribution under the Plan shall comply with the requirements of Section 401(a)(9) of the Code and regulations thereunder, including the minimum distribution incidental benefit requirements of Section 401(a)(9)(G) of the Code.

7.5 Applicability. Sections 7.1 through 7.4 shall apply to all

Members, including Members who had a Separation from Service or ceased to be Eligible Employees prior to January 1, 1989.

- 7.6 Transfer of Account to Other Plan.
- (a) Effective January 1, 1993, if directed by (1) a Participant who is entitled to receive a distribution from the Plan, either

pursuant to this Article or pursuant to Article VI, or (2) the spouse or former spouse of a Participant who is entitled to receive a distribution from the Plan pursuant to a qualified domestic relations order, the Administrator shall direct the Trustee to transfer all or a portion (not Less than \$500) of the amount to be distributed directly to:

- (1) an individual retirement account described in section $408\,(a)$ of the Code,
- $\,$ (2) an individual retirement annuity described in section 408(b) of the Code (other than an endowment contract),
- (3) a qualified defined contribution retirement plan described in section 401(a) of the Code the terms of which permit the acceptance of rollover contributions, or
 - (4) an annuity plan described in section 403(a).
- (b) In addition, if directed by a Participant's surviving spouse who is entitled to receive a distribution from the Plan under Section 7.2, the Administrator shall direct the Trustee to transfer all or a portion (not less than \$500) of the amount to be distributed directly to:
- $$\rm (1)$$ an individual retirement account described in section 408(a) of the Code, or

- (2) an individual retirement annuity described in section 408(b) of the Code (other than an endowment contract).
- (c) The Participant, spouse or former spouse must specify the name of the plan to which the Participant, spouse or former spouse wishes to have the amount transferred, plus such other information as may be requested by the Administrator, on a form and in a manner prescribed by the Administrator.
- (d) Subsections (a) and (b) shall not apply to the following distributions:
- (1) any distribution which is one of a series of substantially equal payments (not less frequently than annually) over either (1) a period of 10 years or more, or (2) a period equal to the life or life expectancy of the Participant or the joint lives or joint life expectancy of the Participant and his beneficiary,
- (2) any distribution if the total distributions paid or payable from the Plan to the same individual during the same calendar year are reasonably expected by the Administrator to be less than \$200,
 - (3) that portion of any distribution after the

Participant's Required Beginning Date that is required to be distributed to the Participant by the minimum distribution rules of section 401(a)(9) of the Code, or

53 08/13/96

- (4) such other distributions as may be exempted by applicable statute or regulation from the requirements of section 401(a)(31) of the Code.
- (e) The transfers described in Subsections (a) and (b) may be effected by the Trustee by the delivery to the Participant (or spouse or former spouse) or Surviving Spouse of a check payable to the trustee of the transferee plan along with instructions addressed to the Participant (or spouse or former spouse) or Surviving Spouse to deliver such check to the trustee of the transferee plan.
 - 7.7 Transfers to Salaried Group. A former Participant in this Plan

who becomes a participant in the P.H. Glatfelter Company 401(k) Savings Plan shall have his Account under this Plan transferred to that plan as soon as practicable following his becoming a participant in that plan. After such transfer, the Employee shall have no further rights under this Plan.

54 08/13/96

SECTION 8

ADMINISTRATION

- 8.1 Administrator. The Administrator of the Plan shall be the
 ----Employee Benefits Committee, a committee of individuals appointed by the Board from time to time.
 - 8.2 Named Fiduciary. The Administrator shall be the named -----

fiduciary responsible for administration of the Plan. The Administrator may, however, by or pursuant to a resolution of its Board of Directors, delegate to any person or entity any of its powers or duties under the Plan. To the extent of any such delegation, the delegate shall become the named fiduciary responsible for administration of the Plan (if the delegate is a fiduciary by reason of the delegation), and references to the Administrator shall apply instead to the delegate. Any action by the Administrator assigning any of its responsibilities to specific persons who are directors, officers, or employees of the Company shall not be a delegation of the Administrator's responsibility but rather shall be treated as the manner in which the Administrator has determined internally to discharge such responsibility.

8.3 Rules of Administration. The Administrator shall adopt such

rules for administration of the Plan as it considers desirable, provided they do not conflict with the express terms of the Plan, and may, in its sole discretion, construe the Plan, correct defects, supply omissions and reconcile inconsistencies to the extent necessary to effectuate the Plan and such action

55 08/13/96

shall be conclusive and binding on all parties. Records of administration of the Plan shall be kept, and Employees and their Beneficiaries may examine records pertaining directly to themselves.

8.4 Services to the Plan. The Administrator may contract for

legal, actuarial, investment advisory, medical, accounting, clerical and other services to carry out the Plan. The costs of such services and other administrative expenses shall be paid in accordance with Sections 10.1 and 10.2 of the Plan.

- 8.5 Funding Policy. The Administrator shall annually review and determine the funding policy of the Plan, with the advice of such experts as it deems appropriate.
 - 8.6 Claims Procedure.
- (a) The Company will advise each Member of any benefits to which he is entitled under the Plan. If any person believes that the Company has failed to advise him of any benefit to which he is entitled, he may file a written claim with the head of the Personnel Department. The head of the Personnel Department shall review such claim and respond thereto within a reasonable time after receiving the claim. The head of the Personnel Department shall provide to every claimant who is denied a claim for benefits written notice setting forth in a manner calculated to be understood by the claimant:
 - (1) the specific reason or reasons for the denial;

- (2) specific reference to pertinent plan provisions on which the denial is based;
- (3) a description of any additional material or information necessary for the claimant to perfect the claim and an explanation of why such material or information is necessary;
 - (4) an explanation of the claim review procedure set

forth in Paragraph (b) below.

(b) Within sixty days of receipt by a claimant of a notice denying a claim under Paragraph (a), the claimant or his duly authorized representative may request in writing a full and fair review of the claim by the Company's Profit Sharing and Stock Ownership Plan Committee. The Company may extend the sixty-day period where the nature of the benefit involved or other attendant circumstances make such extension appropriate. In connection with such review, the claimant or his duly authorized representative may review pertinent documents and may submit issues and comments in writing. The said Committee shall make a decision promptly, and not later than 60 days after the Plan's receipt of a request for review, unless special circumstances (such as the need to hold a hearing, if the said Committee deemed one necessary) require an extension of time for processing, in which case a decision shall be rendered as soon as possible, but not later than 120 days after receipt of a request for review. The decision on review shall be in writing and shall include specific reasons for the decision, written in a manner

57 08/13/96

calculated to be understood by the claimant, and specific references to the pertinent Plan provisions on which the decision is based.

8.7 Liability for Administration. Neither the Company nor any of

its directors, officers or employees shall be liable for any loss due to its or his error or omission in administration of the Plan unless the loss is due to the gross negligence or willful misconduct of the party to be charged or is due to the failure of the party to be charged to exercise a fiduciary responsibility with the care, skill, prudence, and diligence under the circumstances then prevailing that a prudent man acting in a like capacity and familiar with such matters would use in the conduct of an enterprise of a like character and with like aims.

58 08/31/96

SECTION 9

AMENDMENT AND TERMINATION

9.1 Amendment. This Plan may be amended at any time by action of

the Board. It may also be suspended or terminated at any time by the Company, by action of the Board. No amendment, suspension, or termination of the Plan shall divest a Member's vested interest or have the effect of providing that the Fund or any part thereof may be used for or diverted to purposes other than the exclusive benefit of Members or their Beneficiaries.

08/13/96

59

SECTION 10

GENERAL PROVISIONS

- 10.2 Other Expenses. All other costs and expenses, including the
 ----fee of the Trustee and all expenses incurred in the administration of the Plan, shall be borne by the Company.

Annual Statement. The Administrator shall, as soon as

practicable after the end of each Plan Year, deliver to each Member a statement setting forth his interest in the Plan as of the end of such Plan Year. Other statements of account may be delivered from time to time at the discretion of the Administrator.

10.4

08/13/96

10.6 Payments Due Minors or Incapacitated Persons. If the

Administrator deems any person incapable of receiving any payment to which he is entitled by reason of minority, illness, infirmity or other incapacity, it may direct the Company or the Trustee to make payments directly for the benefit of such person or to any person selected by the Administrator to disburse it whose receipt shall be a complete acquittance therefor. Such payments shall, to the extent thereof, discharge all liability of the Company,

the Administrator, the Trustee and the Fund.

10.7 Uniform Administration. Whenever in the administration of the

Plan any action of the Administrator is required with respect to eligibility or classification of Employees, allocations or distributions to Members, or contributions or benefits, such action shall be uniform in nature as applied to all persons similarly situated and no such action shall be taken which will discriminate in favor of Employees who are shareholders, officers or highly compensated Employees.

10.8 Return of Nondeductible or Erroneous Contribution. All

contributions to the Plan are expressly conditioned on their current deductibility by the Company under the Code. If the Internal Revenue Service determines that all or any portion of a contribution is not deductible under Section 404(a) of the Code, the amount determined to be nondeductible shall be returned to the Company if the Company so directs, provided that such repayment may only be made within one year of the disallowance of the deduction. Also, a contribution made by the Company under a

61 08/13/96

mistake of fact shall be returned to the Company within one year after the payment of the contribution if the Company so directs.

10.9 Source of Payment. Benefits under this Plan shall be payable

only out of the Fund and the Company shall have no legal obligation, responsibility, or liability to make any direct payment of benefits accrued under the Plan. Neither the Company nor the Trustee guarantees the Fund against any loss or depreciation or guarantees the payment of any benefit thereunder. No person shall have any right under the Plan with respect to the Fund or against the Trustee or Company except as specifically provided for herein.

Company or to interfere with the rights of the Company to discharge him at any time.

10.11 Inalienability of Benefits. The rights or interests of a

Member or former Member may not be assigned or hypothecated, except pursuant to a qualified domestic relations order or a Federal tax levy made pursuant to Section 6331 of the Code. The Administrator shall review any domestic relations order to determine whether it is qualified within the meaning of Section 414(p) of the Code. An order shall not be qualified unless it complies with all applicable provisions of the Plan concerning mode of payment and manner of elections. Notwithstanding the preceding sentence and any restrictions on

62 08/13/96

may provide for distribution at any time permitted under Section 414(p) (10) of the Code.

10.12 Mergers, etc. No merger or consolidation with, or transfer of

any of the Plan's assets or liabilities to, any other plan shall occur at any time unless each Member would (if the Plan had then terminated) receive a benefit immediately after the merger, consolidation, or transfer which is equal to or greater than the benefit he would have been entitled to receive immediately before the merger, consolidation, or transfer (if the Plan had then terminated).

- 10.13 Governing Law. Except as required by federal law, all aspects -----of this Plan shall be governed by Pennsylvania Law.
 - 10.14 Information Required of Participants. Each individual entitled

to benefits under the Plan shall file with the Company from time to time in writing his post office address and each change of post office address, and any communication, statement or notice addressed to such individual at his last post office address filed with the Company will be binding upon him for all purposes of the Plan. Each individual entitled to benefits under the Plan also shall furnish the Administrator with such documents, evidence, data or information as the Administrator considers necessary or desirable for the purpose of administering the Plan, or to protect the Trustee, the Company or the Administrator.

63 08/13/96

the whereabouts of any individual entitled to benefits under the Plan. Any individual entitled to benefits under the Plan who does not file a timely claim for his benefits will be allowed to file a claim at a later date, and payment of his benefits will commence at that later date.

10.16 Effective Date. Except as otherwise provided herein, the

foregoing provisions of this amended and restated Plan apply only to individuals who are Employees on or after January 1, 1989. The rights of any other individual or beneficiary of a deceased individual shall be governed by the Plan as in effect at the time of such individual's Separation from Service, except to the extent expressly provided in any amendment adopted subsequent thereto.

Executed as of the 28th day of June, 1995.

By: /s/Robert S. Wood
----Robert S. Wood
Secretary

64

08/13/96

AMENDMENT NO. 1

TO THE

PROFIT SHARING PLAN OF
P. H. GLATFELTER COMPANY

AND

THE GLATFELTER PULP WOOD COMPANY
(SPRING GROVE HOURLY GROUP)
(As amended effective January 1, 1989
Restated effective October 1, 1995)

Pursuant to Section 9.1 of the Profit Sharing Plan of P. H. Glatfelter Company and The Glatfelter Pulp Wood Company (Spring Grove Hourly Group), as amended effective January 1, 1989 and restated effective October 1, 1995 (hereinafter, the "Plan"), the Plan is hereby amended in the following particulars to reflect the addition of an investment option in P. H. Glatfelter Company Common Stock, effective September 16, 1996:

1. Section 1.1 is amended by the addition of a new defined term, to read as follows:

"Investment Medium" or "Investment Portfolio" means any fund, contract, obligation, or other mode of investment to which a Participant may direct the investment of the assets of his Account, as designated by the Administrator.

- 2. Section 5.2 is amended in its entirety to read as follows:
- 5.2 Investments and Voting of Shares.
- (a) Except as provided in Subsection (e) of this Section, the Trustee shall invest Elective Contributions and Nonelective Contributions (if any) paid to it and income thereon in such Investment Media, including shares of stock of the Company, as each Participant may select in accordance with this Section. Such investments acquired in the manner prescribed by the Plan shall be held by or for the Trustee.
 - (b) Except as provided in Subsection (e) of this Section, a

Participant shall select one or more of the Investment Media in which his Accounts shall be invested, and the percentage thereof that shall be invested in each Investment Medium selected. If a Participant fails to make an election pursuant to this Section, amounts allocated to his Account shall be invested in the most

conservative of the Investment Media as determined by the Administrator. A Participant may amend such selection by giving prior notice to the Trustee, in a form acceptable to the Trustee and according to procedures determined by the Administrator. Such amendments will be subject to the other requirements of this Section.

- (c) The Trustee shall vote shares of stock of the Company pursuant to instructions from Participants, for which purpose the Trustee shall aggregate fractional shares and vote them in proportion to instructions from Participants with respect to whole shares. Shares for which no instructions are received shall not be voted.
- (d) The amounts contributed by all Participants to each Investment Medium shall be commingled for investment purposes.
- (e) The Trustee may hold assets of the Fund and make distributions therefrom in the form of cash without liability for interest, if for administrative purposes it becomes necessary or practical to do so.
- 3. Section 7.1 is amended by the addition of a new subsection (e), to read as follows:
 - (e) (1) Except as described in paragraph (2) below, all distributions from the Plan shall be made in cash.
 - (2) Notwithstanding the foregoing, a Member or Beneficiary who, as of the date of distribution, has invested all or part of his Account in a stock fund which holds shares of the Common Stock of P.H. Glatfelter Company ("Company Shares") may elect, in accordance with procedures established by the Administrator, to receive distribution of such Company Shares in lieu of a cash distribution. Any such distribution of Company Shares shall be made subject to applicable tax withholding rules.
- 4. Section 10.3 is amended by the addition of the clause "Subject to subsection 5.2(c)," to the beginning thereof.

2 08/13/96

IN WITNESS WHEREOF, based on action by the Company's Board of Directors, this Amendment is executed this 19th day of August, 1996.

BY:/s/Robert S. Wood
----Robert S. Wood
Secretary

EXHIBIT 5.1

September 16, 1996

P. H. Glatfelter Company 228 S. Main Street Spring Grove, PA 17362

RE: PROFIT SHARING PLAN OF P. H. GLATFELTER COMPANY

AND THE GLATFELTER PULP WOOD COMPANY (SPRING GROVE HOURLY

GROUP), REGISTRATION STATEMENT ON FORM S-8

Gentlemen:

We have acted as counsel to P. H. Glatfelter Company (the "Company") in connection with the registration under the Securities Act of 1933, as amended, of interests in the Profit Sharing Plan of P. H. Glatfelter Company and The Glatfelter Pulp Wood Company (Spring Grove Hourly Group), as amended effective January 1, 1989 and restated effective October 1, 1995, and as to be amended by Amendment No. 1 effective September 16, 1996 (the "Plan") and 725,000 shares of common stock of the Company, par value \$.01 per share (the "Shares"), issuable thereunder.

In rendering our opinion, we have reviewed the Plan and such certificates, documents, corporate records and other instruments as in our judgement are necessary or appropriate to enable us to render the opinion expressed below. In giving this opinion, we are assuming the authenticity of all instruments presented to us as originals, the conformity with the originals of all instruments presented to us as copies and the genuineness of all signatures.

Based upon the foregoing, we are of the opinion that (i) the Shares, when issued in accordance with the terms of the Plan, will be legally issued, fully paid and non-assessable, and (ii) the provisions of the Plan satisfy in form the requirements of the Employee Retirement Income Security Act of 1974.

We consent to the filing of this opinion as Exhibit 5.1 to the Registration Statement.

Very truly yours,

/s/ Ballard Spahr Andrews

EXHIBIT 5.2

INTERNAL REVENUE SERVICE DISTRICT DIRECTOR 31 HOPKINS PLAZA BALTIMORE, MD 21201-0000

Date: SEP 18 1995

P.H. GLATFELTER COMPANY C/O BARRY L. KLEIN, ESQUIRE BALLARD SPAHR ANDREWS & INGERSOLL 1735 MARKET STREET PHILADELPHIA, PA 19103-7599 DEPARTMENT OF THE TREASURY

Employer Identification Number: 23-0628360

File Folder Number:

521018524

Person to Contact:

EP/EO CUSTOMER SERVICE UNIT

Contact Telephone Number:

(410) 962-6058

Plan Name:

PROFIT SHARING PLAN OF P.H.

GLATFELTER COMPANY (SPRING GROVE)

Plan Number: 007

Dear Applicant:

We have made a favorable determination on your plan, identified above, based on the information supplied. Please keep this letter in your permanent records.

Continued qualification of the plan under its present form will depend on its effect in operation. (See section 1.401-1(b)(3) of the Income Tax Regulations.) We will review the status of the plan in operation periodically.

The enclosed document explains the significance of this favorable determination letter, points out some features that may effect the qualified status of your employee retirement plan, and provides information on the reporting requirements for your plan. It also describes some events that automatically nullify it. It is very important that you read the publication.

This letter relates only to the status of your plan under the Internal Revenue Code. It is not a determination regarding the effect of other federal or local statutes.

This determination letter is applicable for the amendment(s) adopted on December 22, 1994.

This plan has been mandatorily disaggregated, permissively aggregated, or restructured to satisfy the nondiscrimination requirements.

At your request, this determination letter does not express an opinion, and may not be relied on with respect to, whether the nondiscrimination in amount requirement of section 1.401(a)(4)-1(b)(2) of the regulations has been

satisfied.

This letter is issued under Rev. Proc. 93-39 and considers the amendments required by the Tax Reform Act of 1986 except as otherwise specified in this letter.

This plan satisfies the nondiscriminatory current availability requirements of section 1.401(a)(4)-4(b) of the regulations with respect to those benefits, rights, and features that are currently available to all employees in the plan's coverage group. For this purpose, the plan's coverage group consists of those employees treated as currently benefiting for purposes of

-2-

P.H. GLATFELTER COMPANY

demonstrating that the plan satisfies the minimum coverage requirements of section 410(b) of the Code.

This letter may not be relied upon with respect to whether the plan satisfies the qualification requirements as amended by the Uruguay Round Agreements Act, Pub. L. 103-465.

The information on the enclosed addendum is an integral part of this determination. Please be sure to read and keep it with this letter.

We have sent a copy of this letter to your representative as indicated in the power of attorney.

If you have questions concerning this matter, please contract the person whose name and telephone number are shown above.

Sincerely yours,

/s/ Paul M. Harrington

District Director

Enclosure(s)
Publication 794
Addendum

-3-

P.H GLATFEELTER

This determination letter includes the following Controlled Group Employers; P.H Glatfelter Company, The Glatfelter Pulp Weed Company and Spring

Grove Water Company.

EXHIBIT 5.3

The opinion of Ballard Spahr Andrews & Ingersoll concerning compliance with the requirements of ERISA is contained in its opinion filed as Exhibit 5.1 to the Registration Statement

EXHIBIT 15

LETTER IN LIEU OF CONSENT REGARDING REVIEW OF REPORT OF FINANCIAL INFORMATION

P.H. Glatfelter Company:

We have made a review, in accordance with standards established by the American Institute of Certified Public Accountants, of the unaudited condensed consolidated financial statements of P.H. Glatfelter Company and subsidiaries for the periods ended March 31, 1996 and 1995, and June 30, 1996 and 1995, as indicated in our reports dated April 15, 1996 and July 16, 1996; because we did not perform an audit, we expressed no opinion on that information.

We are aware that our reports referred to above, which were included in your Quarterly Report on Form 10-Q for the quarters ended March 31, 1996 and June 30, 1996 are being used in the Registration Statement on Form S-8.

We also are aware that the aforementioned report, pursuant to Rule 436(c) under the Securities Act of 1933, is not considered a part of the Registration Statement prepared or certified by an accountant or a report prepared or certified by an accountant within the meaning of Sections 7 and 11 of that Act.

DELOITTE & TOUCHE LLP Philadelphia, Pennsylvania September 9, 1996

(#8476)

EXHIBIT 23.1

INDEPENDENT AUDITORS' CONSENT

We consent to the incorporation by reference in this Registration Statement of P.H. Glatfelter Company on Form S-8 of our report dated February 2, 1996, appearing in and incorporated by reference in the Annual Report on Form 10-K of P.H. Glatfelter Company and subsidiaries for the year ended December 31, 1995.

DELOITTE & TOUCHE LLP Philadelphia, Pennsylvania September 9, 1996

EXHIBIT 23.2

CONSENT OF COUNSEL

The consent of Ballard Spahr Andrews & Ingersoll is contained in its opinion filed as Exhibit 5.1 to the Registration Statement.